Author's response to reviews

Title: Building the capacity of family day care educators to promote children's social and emotional wellbeing: An exploratory cluster randomised controlled trial

Authors:

Elise Davis (eda@unimelb.edu.au)
Lara Williamson (wlm@unimelb.edu.au)
Andrew Mackinnon (andrew.mackinnon@unimelb.edu.au)
Kay Cook (kay.cook@deakin.edu.au)
Elizabeth Waters (ewaters@unimelb.edu.au)
Helen Herrman (h.herrman@unimelb.edu.au)
Margaret Sims (m.sims@une.edu.au)
Cathrine Mihalopoulos (cathy.mihalopoulos@deakin.edu.au)
Linda Harrison (lharrison@csu.edu.au)
Bernie Marshall (bernie.marshall@deakin.edu.au)

Version: 2 Date: 5 October 2011

Author's response to reviews:

We have amended our manuscript to indicate that we have received ethics approval.